Author's response to reviews

Title: Treatment compliance and effectiveness of a cognitive behavioural intervention for low back pain: A complier average causal effect approach to the BeST data set.

Authors:

Christopher R Knox (C.R.H.Knox@warwick.ac.uk)
Ranjit Lall (R.Lall@warwick.ac.uk)
Zara Hansen (z.hansen@warwick.ac.uk)
Sarah E Lamb (S.Lamb@warwick.ac.uk)

Version: 2 Date: 13 June 2013

Author's response to reviews:

An acknowledgments section has been added and the trial number revised as requested.